Author's response to reviews

Title: Novel polymorphisms in caspase-8 are associated with breast cancer risk in the California Teachers Study: a nested case control study

Authors:

Hannah L Park (hlpark@uci.edu)
Argyrios Ziogas (aziogas@uci.edu)
Jenny Chang (jichang@uci.edu)
Bhumi Desai (desaib13@gmail.com)
Leona Bessonova (leona.bessonova@live.com)
Chad Garner (cgarner@uci.edu)
Eunjung Lee (Eunjung.Lee@med.usc.edu)
Susan L Neuhausen (sneuhausen@coh.org)
Sophia S Wang (sowang@coh.org)
Huiyan Ma (HMa@coh.org)
Jessica Clague (jclague@coh.org)
Peggy Reynolds (Peggy.Reynolds@cpic.org)
James V Lacey Jr (jlacey@coh.org)
Leslie Bernstein (LBernstein@coh.org)
Hoda Anton-Culver (hantoncu@uci.edu)

Version: 2
Date: 25 March 2015

Author's response to reviews:

Dear Editors,

Thank you for your editorial assessment. We have made the following fixes:

1) Line numbering: Line numbers were added to the entire document.

2) Requesting Ethics statement: A subsection was added to the Methods section called “Ethics statement” (lines 90-96).

Please let me know if any other information is needed. Thank you.

Sincerely,
Hannah Lui Park, Ph.D.